Author's response to reviews

Title: The health disparities cancer collaborative: a case study of practice registry measurement in a quality improvement collaborative

Authors:

David A Haggstrom (dahaggst@iupui.edu)
Steven B Clauser (clausers@mail.nih.gov)
Stephen H Taplin (taplins@mail.nih.gov)

Version: 5 Date: 13 May 2010

Author's response to reviews:

p. 13, 2nd sentence, "experience" has been changed to "measurement"

p. 14, 2nd paragraph, last sentence, deleted "then"

p. 14, 1st paragraph, 2nd to last sentence, added "the measures and comparisons of"

p. 19, 2nd paragraph, last sentence, moved "complete registries will be difficult to implement" to beginning of sentence

a new reference 20 was added and the remaining references were renumbered accordingly

I have ensured that the title is identical in the manuscript and the submission system, using sentence case.

I have ensured that the list of authors in the manuscript is written exactly as they are in the submission system, both in style and order.

I have removed the visible vertical lines from the tables.

I have not used ranges when listing figures, but listed all figures.

I have removed the titles from the figures, and placed the figure titles after the reference list.

For the final layout of the manuscript, I have cropped the figures as closely as possible using details for preparing files for submission from here: http://www.biomedcentral.com/info/ifora/figures